Author's response to reviews

Title: Wnt/beta-Catenin Pathway Regulates ABCB1 Transcription in Chronic Myeloid Leukemia.

Authors:

Stephany Corrêa (scorrea@biof.ufrj.br)
Renata Binato (rbgomes@inca.gov.br)
Bárbara Du Rocher (barbaradurocher@inca.gov.br)
Luciana Pizzatti (pizzatti@inca.gov.br)
Eliana Abdelhay (eabdelhay@inca.gov.br)

Version: 2 Date: 4 November 2011

Author's response to reviews: see over
To Drª. Jigisha Patel, Series Editor of the “BMC Cancer Journal”

Dear Drª. Patel,

We are re-submitting the manuscript titled, “Wnt/β-Catenin Pathway Regulates $ABCB1$ Transcription in Chronic Myeloid Leukemia,” for consideration of publication in the “BMC Cancer Journal”. The authors made the changes requested by the Editorial office as follow:

- In section **METHODS**, sub-section **Bone marrow samples**, the approval of ethics committee board, with its description, and patients consent for participation in the study were added properly.

Please let me know if I successfully addressed all queries requested. If any more changes are needed, I will be pleased to try to correspond.

Hope to hear from you soon.

Best regards,

Stephany Corrêa.